Oxford Handbooks Online

The Illegal Wildlife Trade a

Stephen F. Pires and William D. Moreto

Subject: Criminology and Criminal Justice, International and Comparative Criminology,

Organizational and White-Collar Crime

Online Publication Date: Jul 2016 DOI: 10.1093/oxfordhb/9780199935383.013.161

Abstract and Keywords

The illegal wildlife trade is a growing problem driven by a number of factors (e.g. subsistence, alternative medicine, accessories, the pet trade). High demand for illicit wildlife products is threatening the existence of many of the most-endangered species. By unsustainably removing coveted species from the wild, communities that depend on such species for subsistence or eco-tourism will be adversely impacted by depleting populations. Laws and regulations have been implemented over the years, most notably CITES, to regulate the commercial trade in wildlife and prohibit trade in other species that are at-risk of overexploitation albeit with mixed success. Criminologists have recently entered the fold and provided insight to the wildlife trade through various perspectives. Researchers are beginning to better understand why and how the trade operates and what solutions might be implemented to reduce it. The article ends with implications for future research.

Keywords: wildlife crime, environmental criminology, green criminology, conservation criminology, poach, trafficking

Introduction

The illegal wildlife trade has long been a problem that only recently has captured the attention of the public and academics across disciplines. Criminologists and criminal justice scholars are increasingly researching the topic from a variety of perspectives with the aim of understanding the nature of the trade and how to mitigate it. Section I discusses the nature of the illegal wildlife trade and its impact on fauna and flora. An indepth examination of international agreements and regulations is the focus of Section II. Section III examines the theoretical perspectives to understanding the illegal trade in wildlife. The types of offenders involved and the organization of the trade is detailed in Section IV. Section V identifies supply markets, or where poaching disproportionately occurs, along with where major demand markets are found. Section VI identifies the reasons why wildlife is trafficked and covers three commonly poached animals in the illegal wildlife trade—rhinos, pangolins, and parrots—in greater depth. The final section offers implications for future research.

I. Nature and History of the Problem

According to Interpol (2015), "Wildlife crime is the taking, trading, exploiting or possessing of the world's wild flora and fauna in contravention of national and international laws." Whether species are poached for subsistence (e.g., bushmeat) or personal ownership (e.g., ivory bangles), for local or international trade, cultural or religious beliefs, or as a result of human-animal conflict, a wildlife crime has occurred. Such a definition, however, does not cover "harms" to animals that may lead to diminished populations but is otherwise legal (also see Section III; White 2008). Such harms commonly come in the form of habitat loss (or deforestation), the practice of clearing a large number of trees for agricultural purposes or for human settlement (Geist and Lambin 2002). Deforestation is the primary threat of extinction for the majority of wildlife on earth because it removes suitable habitat for millions of species (Pimm and Raven 2000) and exacerbates climate change (MEA 2005). Because deforestation is commonly sanctioned by governments, it is not a crime in the traditional sense and will not be the focus of this systematic review.

For centuries, wildlife was taken from the wild for profit, personal use, or killed because they were a nuisance. Retaliatory killing of elephants, tigers, and lions, for example, has been a common practice in range states after attacking humans and/or their livestock (Omondi et al. 2004). Many of these actions were not against the law. Even in times or places where it was unlawful, it was often overlooked or ignored by law enforcement. In cases where animals were trapped or poached for profit, much of this trade was for local markets. As the world became more globalized in the latter decades of the 20th century, so too did the illegal wildlife trade. Species once poached for local or national demand are now trafficked through multiple countries and ports to reach markets on the other side of the world. Poachers may no longer be exclusively locals; outsiders are now sometimes involved in search of highly endangered and valuable species such as rhinos and elephants. Because of this international demand in combination with accelerated habitat loss, the illegal wildlife trade is significantly contributing to the decline of many species (Broad et al. 2003, Butchart et al 2010; Phelps et al. 2010; Sutherland et al. 2009), and in particular, a significant proportion of endangered species (Rosser and Mainka 2002). Some have claimed that we are currently experiencing the sixth mass extinction, where up to 100,000 species go extinct on an annual basis due to humanbased activity (Barnosky et al. 2011; Ceballos et al. 2015).

Assessing the extent of the illegal wildlife trade is made difficult due to the clandestine nature of illicit activity. Economic estimates suggest the illegal wildlife trade, excluding the illegal timber trade and illegal fishing, is valued between US\$7.8 and 10 billion a year (Haken 2011). The illegal fishing industry is estimated to be between US\$10 and 23 billion a year (Agnew et al. 2008), and illegal timber is valued at US\$7 billion a year (Haken 2011). Estimates such as these make the illegal wildlife trade one of the most

Page 3 of 41

PRINTED FROM OXFORD HANDBOOKS ONLINE (www.oxfordhandbooks.com). (c) Oxford University Press, 2015. All Rights Reserved. Under the terms of the licence agreement, an individual user may print out a PDF of a single chapter of a title in Oxford Handbooks Online for personal use.

Subscriber: University College London; date: 05 May 2018

profitable illicit trafficking industries behind the trafficking of drugs, guns, and humans (Vince 2002).

The illicit trade has severe repercussions for both humans and wildlife. Unregulated taking of natural resources depletes resources for nations and the local communities that depend on them for their livelihoods (TRAFFIC 2008). The illegal wildlife trade also happens to have a high mortality rate due to the clandestine nature of trafficking live species (Cantu et al. 2007; Clark, Van Thai, and Phuong 2008; Franke and Telecky 2001). Often poachers take more than what isnecessary as there is an expectation that many wildlife individuals will perish en route to markets. Apart from the economic and population losses, poachers also damage the environments of species by using destructive practices such as felling trees (Gonzalez 2003), using cyanide to stun fish, and employing "blast fishing" (i.e., use dynamite) that causes coral degradation (McClellan et al. 2008).

Since at least the 1970s, there has been growing consensus that the wildlife crime problem needs to be addressed in the form of new policies, regulations, protected parks, enforcement, and public education. In addition to this, conservationist organizations, such as TRAFFIC, the World Wide Fund for Nature (WWF), and the International Union for Conservation of Nature (IUCN), have been at the forefront of protecting endangered species and their habitats while studying the global illegal wildlife trade problem in greater depth (see Schneider 2012). The following section examines regulations and laws put forth at the international and national levels.

II. Regulations and Laws

A. International Regulations

Despite the best intentions and efforts, combating the illegal wildlife trade effectively cannot fall upon a single nation in the world. Most wildlife is not endemic to one country, and wildlife often migrates across geopolitical boundaries. Criminals in the trade often facilitate this migration by poaching, trafficking, processing, and selling wildlife to consumers that span several countries and regions of the world. For these reasons, combating the illegal wildlife trade must be a coordinated effort that utilizes local, national, regional, and international cooperation (Schneider 2012; Pires and Moreto 2011; Moreto 2015).

A number of bilateral and multinational agreements have been put in place over the last century or so to conserve species in danger of extinction (Mitchell 2003; Schneider 2012). In the first half of the 20th century, such agreements, in addition to national legislation, restricted hunting and trading of wildlife in various ways. Many of these agreements and national regulations were unsuccessful in their objectives, which allowed unsustainable

Page 4 of 41

wildlife poaching and trafficking to persist. For example, the International Agreement for the Regulation of Whaling signed in 1937 was seen as a failure because it did not impose size or catch limits (Schneider 2012).

The failure of national regulations and international agreements, along with the rise in the illegal wildlife trade of threatened species, instigated the 1973 drafting of the Convention on International Trade in Endangered Species of Wild Fauna and Flora (CITES). This was subsequently put into effect in 1975 with 80 nations agreeing to regulate trade in wildlife and prohibit the trade of the most at-risk species from overexploitation. Since then, 181 nations have become part of the CITES agreement and have adopted CITES policies at the national level (Moyle 2003) through a licensing system that subjects certain wildlife to varying degrees of controls. To this day, CITES is considered to be the most effective international agreement in history in regulating the wildlife trade (Phelps et al. 2010; Huxley 2000; Schneider 2012).

CITES introduced three appendices in which listed species are afforded greater protection when traded between countries (see Table 1). Appendix I species are prohibited in international commercial trade because the trade may greatly impact these vulnerable species. The African elephant, cheetah, giant panda, tigers, and most rhino species are among the most notable species listed under Appendix I. Appendix II species are not always at-risk of extinction but can be at-risk without a system of regulations. Such species may be commercially traded only if export permits are procured, and depending on the import nation, may also require import permits. Appendix III is reserved for species that are requested to be listed and protected by a member nation so that other member nations avoid exploitation of said species (CITES 2015a).

Table 1 CITES Appendices and Restrictions on Trade			
CITES Appendices	Restrictions on Trade	Number of Species Covered	
CITES I	Species afforded greater protection because they are threatened with extinction. Species cannot be traded for commercial purposes. If they are traded, both import and export permits are required that demonstrate the trade "is not detrimental to their survival."*	931	
CITES II	Species may or may not be threatened with extinction, but the trade of such species needs to be highly regulated. Most species can be commercially traded with a required export permit.	34,419	

Page 5 of 41

CITES III	Species are protected in at least one country, and	147
	this country or set of countries has asked others to help regulate the trade to avoid overexploitation.	

(*) According to CITES 2015c.

(**) CITES, 2015b.

The effectiveness of CITES trade controls on saving species from extinction is mixed. Many claim CITES is highly effective (e.g., Cole 2012; Fuchs 2010; Huxley 2000), but this is difficult to conclude in the absence of a systematic study examining the impact of listing species under CITES (Weber et al. 2015; Roe 2008; Martin 2000). The evidence available is restricted to species-specific studies and the impact CITES trade controls have at the macrolevel. From this, evidence has shown that trade bans, that is, listing species under Appendix I, have resulted in less trade and rebounding populations in some species such as the African elephant, vicunas, the southern white rhino, and the greater one-horned rhino (WWF 2015; Eloff and Lemieux 2014; Cole 2012; Lemieux and Clarke 2009; Mcallister et al. 2009). Often this success is in combination with conservation efforts on the ground (WWF 2015; Eloff and Lemieux 2014), and where it includes a recovery plan that calls for consultation with local communities and range states (Larriera et al. 2010; Lichtenstein 2009; Frisina and Tareen 2009; Mcallister et al. 2009; Hutton and Webb 2003; Leader-Williams 2003).

Critics of CITES actions suggest that declines are more common than recoveries for species listed under CITES ('t Sas-Rolfes 2010, 2012; Dickson 2003; Kievit 2000; Martin 2000; ERM 1996). Even when utilizing trade bans, populations do not automatically recover once commercial trade and poaching cease (Leader-Williams 2003; Jachmann 2003). A number of other factors, such as habitat loss and human-animal conflict, can impact population recoveries independent of the reduction in the illegal wildlife trade. In addition to these critiques, the complexity of demand and market dynamics is not always considered by CITES policymakers when implementing trade bans (Challender et al. 2015a; 2015b; Rosen and Smith 2010; Hall et al. 2008). This suggests trade bans, the most important tool CITES has in its arsenal to combat the illegal wildlife trade, may be an overly simplistic response to a very complex problem (Bowman 2013; Briggs et al. 2013; Couzens 2013; Moore 2011a; Cooney and Jepson 2006; Dickson 2003; Moyle 2003). Because of the possible overuse of trade bans in the absence of viable conservation policies on the ground, endangered species may not recover from dwindling populations, which has been the case with tigers (Project Tiger 2005) and pangolins (Pantel and Chin 2009) in particular. Overall, trade bans have the potential to be a useful tool in the shortterm but may lead to poaching and wildlife trafficking in the long-term if no other strategies are implemented (Hutton and Webb 2003; Conrad 2012).

CITES has the ability to save some species from extinction and protect others from being at-risk, however, the success of CITES is contingent on a number of working parts that may be lacking or failing altogether. First, the lack of enforcement and compliance of CITES objectives at the national and local levels has hindered its overall effectiveness (Han 2014; Schneider 2012; Rosen and Smith 2010; Leader-Williams 2003; Vasquez 2003). This lack of enforcement and compliance disproportionately occurs on the continent of Africa where many of the most biodiverse regions of the world are. Noncompliance and enforcement problems in Africa are due to a myriad of problems that include (1) corruption, (2) wars, (3) limited resources, (4) lack of political commitment and stability, and (5) domestic conflict (Schneider 2012; Fiadjoe 2004). Second, evaluating whether CITES policies are effective is difficult given data limitations. For example, population counts are not collected for many threatened species that are listed under CITES (Abensperg-Traun et al. 2009; Phelps et al. 2010; Parsons et al. 2010). Third, trade bans may increase black market prices for threatened and endangered species, which can exacerbate poaching (MacMillan and Han 2011; Pires and Moreto 2011; Hall et al. 2008; Courchamp et al. 2006; 't Sas-Rolfes 2000). Fourth, locals may not be incentivized to preserve wildlife when the ability to legally trade them is removed. Some suggest that people only value and protect wildlife if they can benefit from using it (Dickson 2003; Leader-Williams 2003; Martin 2000). Finally, CITES is limited because it is designed to address only the illegal international trade in wildlife and not intranational trade. Recent research shows the domestic trade in illegal wildlife is guite substantial (Pires 2012; Pires and Moreto 2011; Nijman 2010; Tilson et al. 2010; Du Plessis 2000) and may even be larger than the international trade for some species (see also Section VI).

B. National Regulations

Typically nations will either impose poaching (and trade) bans for all species of a particular animal or regulate the trade in the form of catch-quotas. Catch-quotas allow a certain number of species to be trapped every year as long as appropriate permits have been obtained. The lobster industry, for example, illustrates this scheme quite well: fishermen are required to obtain permits and/or licenses to trap lobsters during particular times of the year and are restricted by catch limits and size. The purpose of such a regulatory scheme is to allow certain members of the community to profit from historical practices while ensuring species' populations are healthy. Ideally, catch-quotas are based on scientific assessments of populations so that legal trapping, or catches, of wildlife are sustainably based, but this may not always be the case (Cantu et al. 2007).

To date, no systematic study has analyzed the effects of either poaching bans or catch-quota regulations on species' populations (Smith and Walpole 2005). One recent study suggests that export quotas of birds had no influence on the domestic bird trade in a neotropical country (Daut et al. 2015). However, other parrot-trade-related studies have found catch-quotas can lead to overtrapping due to lack of enforcement (Cantu et al. 2007; Gastanaga et al. 2011). There is similarly mixed evidence regarding the effectiveness of poaching bans. Many of the most-endangered and trafficked species have habitats in nations with total poaching and trade bans. Clearly, a systematic review or meta-analysis study is needed to shed light on which approach is better at diminishing the risk of extinction in species at the national level.

III. Theoretical Perspectives to Understanding the Illegal Wildlife Trade

We now shift our attention to discussing the criminological and criminal justice perspectives that have been used to examine illegal wildlife trafficking and related wildlife crime (e.g., poaching). For ease, we focus on the four main perspectives that have dominated the literature: environmental criminology and crime science, green criminology, conservation criminology, and criminal justice. However, it is worth noting that scholars have used other criminological perspectives to examine wildlife crimes that go beyond the following (e.g. techniques of neutralization, differential association theory, etc.; see Eliason 1999; Eliason and Dodder 1999).

A. Environmental Criminology and Crime Science

Although, the term "environmental criminology" has also been used to specifically describe the study of crimes against the environment (White 2008), in order to avoid confusion (see Brisman and South 2013; Herbig 2014), we adhere to the original and traditional conceptualization and use of environmental criminology. With its roots in crime prevention (Andresen 2014), environmental criminology is made up of established theoretical frameworks, including the routine activity approach, geometric theory of crime, rational choice perspective, and pattern theory. In a similar vein, crime science is a multidisciplinary applied action-research perspective that is outcome focused and based on understanding the "how" of crime as opposed to the "why" in order to develop preventative strategies for immediate crime reduction (Clarke 2004; Laycock 2005).

Research over the years has consistently shown crime disproportionately occurs as it relates to target selection (Cornish and Clarke 1986), space (Sherman et al. 1989), time (Felson and Poulsen, 2003), and products (Clarke 1999). By knowing how crime is concentrated, opportunity-mitigating strategies can be implemented to have immediate crime reduction effects. Arguably, one of the most widely known crime prevention strategies is situational crime prevention (Clarke 1980, 1997, 2009), which aims to alter the immediate environment for a specific crime type in order to increase the effort and risk for crime commission, while also reducing or removing rewards, provocations, and excuses.

To date, a number of studies have been conducted on the illegal wildlife trade from an environmental criminology and crime science perspective. For example, the routine activities approach (Cohen and Felson 1979) has been used to examine wildlife poaching in protected areas in South Africa (Herbig 2011) and trophy poaching in Montana in the United States (Eliason 2012). Recently, Moreto and Lemieux (2015b) extended the routine activity framework to incorporate the existence and role of proxy offenders (i.e., wire snares initiating contact with wildlife as opposed to poachers) in their research of poaching in Uganda.

Crime scripts, also known as the procedural analysis of offending, which attempts to generate and organize the procedural activities and requirements of crime commission (Cornish, 1994), has been used to examine wildlife trafficking (Lavorgna 2014; although see Moreto and Clarke 2013). Additionally, the concept of "hot products," or products that are "most attractive to thieves" (Clarke 1999: 1), and the related conceptual CRAVED framework (concealable, removable, available, valuable, enjoyable, and disposable) has been applied to several wildlife crimes, including parrot poaching (Pires and Clarke 2011, 2012), parrot trafficking (Pires and Petrossian 2016), illegal fishing (Petrossian and Clarke 2014), and livestock theft (Sidebottom 2013). With respect to the illegal wildlife trade, Moreto and Lemieux (2015a) recently argued that the CRAVED model may be limited in fully assessing the dynamic nature of some wildlife products as they maneuver

Page 9 of 41

through an illegal market continuum. They therefore proposed an extended model referred to as CAPTURED (concealable, available, processable, transferrable, useable, removable, enjoyable, and desirable).

Lastly, crime reduction and prevention frameworks have also been introduced within the scope of wildlife crimes, especially situational crime prevention (Petrossian et al. 2016; Lemieux 2014; Pires and Moreto 2011). For instance, the situational crime prevention framework has been used to investigate the impact of the international ban on ivory sales on elephant poaching in Africa (Lemieux and Clarke 2009) and illegal fishing (Petrossian 2015). Additionally, the market reduction strategy, which was initially conceived to reduce and disrupt stolen goods markets, has also been discussed within the scope of wildlife trafficking (Schneider 2008).

B. Green Criminology

Considered to be within the "margins of criminological research [and a] distinctive subfield that challenges traditional approaches in criminological inquiry" (Wyatt 2013: 13), green criminology originally developed as an extension of radical criminology. In essence, green criminology was proposed as a means to investigate the underlying and overreaching political economic origins of environment-related criminal activity (Stretesky, Long, and Lynch 2013). From this perspective, it is argued that the causes of environmental harms are similar to factors that result in social injustice in that inequalities associated with gender, race, and class result in harms against both humans and the environment. In other words, environmental and social injustice is perpetuated by established power dynamics and by those who control the means of production (White 2008; Stretesky et al. 2013).

While some green criminologists still operate from a political economy perspective, including more recently in relation to the role of the treadmill of production and ecological disorganization on environmental crimes (Lynch et al. 2013; Stretesky et al. 2013), green criminology has been expanded to better account for "environmental justice, with a special focus on human rights and social equity; and ecological justice, with a special focus on the biosphere generally and the rights of non-human as well as human" (White 2008: 50). Notably, most of those who adhere to a green criminological viewpoint argue that solely focusing on criminal violations is too restrictive and that a harm perspective is needed to account for activities that cause significant environmental destruction but may fall outside the purview of criminal law (Lynch and Stretesky 2003; White 2008). As such, it has been argued that the broad scope and purpose of green criminology is to

provide a space within criminology to examine the nexus between environmental problems, the definition of harms against nature as crimes, the need to reconsider criminal justice practices and policies in relationship to the environmental harms

they produce, the variety of victims environmental offenses create ... and the effect of environmental toxins on ecological systems and species' health and behavior.

(Lynch and Stretesky 2014: 2)

Acknowledging the importance of environmental harms, green criminologists also recognize the importance of distinguishing between different levels of eco-justice, including environmental justice (i.e., emphasis on human well-being), ecological justice (i.e., focus on investigating human interaction with the environment as it relates to harms and risks), and species justice (i.e., consideration of the welfare and rights of nonhuman animals) (White 2008; White and Heckenberg 2014). Additionally, green criminologists have also proposed the importance of developing a nonspeciesist perspective (Beirne 1999), exploring the link between cultural meanings and perspectives as it relates to environment-related crimes (Brisman and South 2013) and linkages between environmental harm and racial and socioeconomic inequalities (Brisman 2015). Moreover, the transnational nature of environmental crimes/harms has been recognized within the promotion of an eco-global criminology framework (White 2011). With reference to wildlife trafficking, a green criminological perspective has been used to examine the trade in Vietnam (Ngoc and Wyatt 2013) and in Russia (Wyatt 2009).

C. Conservation Criminology

Partly in response to some of the criticisms and limitations levied against green criminology (see Halsey 2004 for example), Gibbs and colleagues (2010) introduced conservation criminology as an interdisciplinary conceptual framework that integrates criminology, natural science disciplines, and risk and decision science. Similar to green criminology, conservation criminology recognizes the importance of examining behaviors that do not violate criminal law, however, conservation criminologists "reject previous definitions and typologies of conservation crime" that can be found within the green criminological literature that emphasized social justice and biocentric perspectives (Gibbs et al. 2010: 129).

The primary starting point for conservation criminologists is to determine and ascertain the manifestation and level of environmental risk from the perspectives of different stakeholders. By focusing on risk, as opposed to crime for example, it is argued that the inclusion of different perspectives can be attained, potential consequences can be better articulated, and different solutions can be developed. Conservation criminology also operates from multiple scales (e.g., individual to collective) and domains (e.g., local to global) thereby providing a comprehensive examination of environmental risks, both direct and indirect (Gibbs et al. 2010).

In addition to incorporating risk and decision sciences, conservation criminology explicitly integrates the natural science disciplines in order to better understand and respond to environmental crimes. Given the expansive scope of environment-related crimes and the overall lack of familiarity of criminologists or criminal justice scholars in measuring, managing, and conserving natural ecosystems, it is argued that natural science disciplines have a very important role in examining environmental crimes. As put by Gibbs and colleagues (2010):

Criminologists are knowledgeable about the legal system, but are not trained to scientifically assess threats to the natural environment. Thus, criminologists could provide information on the circumstances under which legal versus other tools may be most effective while natural resource scientists offer insight into the impact of various stimuli on natural resources and ecosystems. (133)

By incorporating three distinct disciplines, conservation criminology therefore "offers a model for understanding this type of illicit human behavior (environmental crimes) and the emotions, cognitions, and institutions that affect human relations with the environment" (Gore 2011: 659). Within the scope of wildlife crimes, conservation criminologists have examined poaching risks (Kahler, Roloff, and Gore 2013) and noncompliance with poaching laws (Kahler and Gore 2012) in Namibia.

D. Criminal Justice Perspectives

In addition to criminologists, criminal justice scholars have provided important contributions to the study of wildlife crimes, particularly within the scope of wildlife law enforcement. Not only has this research provided insight into better understand the human dimensions of conservation policy (Jacobson and Duff 1998; Gore 2011) and central aspects of wildlife law enforcement (see Forsyth 1993; Eliason 2003, 2006; Shelley and Crow 2009; Patten 2010; Warchol and Kapla 2012; Moreto 2015; Moreto, Brunson, and Braga 2015, Forthcoming), but such frontline perspectives have been vital for research on wildlife crime as well, including the development of poaching typologies (Forsyth, Gramling, and Wooddell 1998; Eliason 2008). Moreover, such research has provided valuable insight into conditions that may contribute to the development and continuance of wildlife crimes, such as ranger deviance (Moreto et al. 2015).

IV. Actors in the Trade

A. The Market Continuum from an Actor-based Perspective

The number of individuals required for a wildlife product to maneuver through a market will vary and depend on a number of factors, including the expected end market and anticipated consumer, the unique characteristics of the products, and the capabilities and limitations of actors already involved in the trade. While the illegal wildlife trade can be examined in various ways, including a stage-based (Tailby and Gant 2002) or a product-based approach (Moreto and Lemieux 2015a), we focus our attention on the actors in the trade, in part because the emphasis in the literature is on the individuals involved. Indeed, "the journey of any given wildlife product from the collector at the source to the final consumer can involve a wide range of intermediaries and other stakeholders" (Broad et al. 2003: 15). It is important to note that the following categories are not mutually exclusive and that individuals may fulfill multiple roles throughout the trade. Furthermore, we explicitly refer to these individuals as actors as opposed to offenders given the potential conceptual overlap between offender/victim as it pertains to broader societal changes and cultural pressures (i.e., legal hunting deemed illegal; see Wyatt 2013).

B. Poachers

In comparison to other actors involved in the illegal wildlife trade, poachers have received the most empirical attention. Research suggests that poachers vary; they are influenced by different motivations and motives, utilize different techniques, and can use a variety of equipment. As evidenced in the literature, different wildlife species may be targeted for different reasons and therefore, specific techniques may be preferred over others. For example, if a tiger is targeted for its skin, poison may be used instead of

Page 13 of 41

hunting techniques that may damage the pelt (e.g., guns). Broadly speaking, research suggests that poachers are driven to poach because of necessity/subsistence, profit/commercial, traditional or cultural practices, religious beliefs, boredom, protection of self and property, rebellion, thrill killing, trophy hunting, research, zoos, exploitation of legal hunting practices, and gamesmanship (Muth and Bowe 1998; McMullan and Perrier 2002; Raymakers 2002; Tailby and Gant 2002; Zimmerman 2003; Shepherd and Magnus 2004; Bell, Hampshire, and Topalidou 2007; Ahmed 2010; Wilson-Wilde 2010; Wyatt 2013; Moreto and Lemieux 2015b; von Essen et al. 2014).

C. Middlemen/Traders

In wildlife trafficking, Wyatt (2013) argues that middlemen are akin to a "fence [or] someone who moves stolen goods within networks" (87). Middlemen are involved in brokering transactions between different parties, acting as wholesalers, and when necessary, keeping in storage illegal wildlife contraband (Broad, Mulliken, and Roe 2003). Often middlemen are locals (Broad et al. 2003; WWF 2012) who may be trading due to their familiarity with the underground trade or individuals with a vested interest in specific products; however, within the scope of organized crime groups (discussed later), middlemen may simply be an extension of a larger trafficking operation, albeit one that is based more on loose networks rather than a hierarchical one (Wyatt 2013).

D. Processors

As mentioned earlier, prior research suggests that the characteristics of wildlife products should be taken into consideration when investigating the entirety of an illegal trade. For example, a poached tiger can be redistributed via the illegal market in several ways: from a live pet to different products used for traditional Asian medicine (TAM). Both a live tiger and tiger parts will require individuals who are familiar with specialized knowledge and techniques. In other words, wildlife may need to be modified to ensure that it is a "saleable product" (Wyatt 2013: 6), or, more broadly, the "alteration, refinement, or conversion of a product's form or status" (Moreto and Lemieux 2015a: 311) may need to be performed for the product to successfully move along the trade. Notably, the latter definition recognizes that the illegal wildlife trade is not simply based on monetary transactions; it may involve customary or traditional practices and is therefore wider in definition. Moreover, this definition explicitly highlights how the physical form of a product can be changed as well as its legal or regulatory status. The reality that some wildlife products may need to be processed demonstrates how the product also dictates who needs to be involved in the trade. For example, unique skills are required to successfully convert raw ivory into decorative items (Vira et al. 2014) and sturgeon roe into caviar (Saffron 2002).

E. Transporters

Transporters are responsible for transporting the wildlife product to its next destination in the market chain. Given the variability in both domestic and transnational forms of wildlife trafficking, transporters can operate within countries and across borders (Wyatt 2013). At times referred to as "mules" (IFAW 2013), transporters can travel using a variety of transportation modes and use a number of techniques for concealing wildlife products. For example, transporters have attempted to smuggle wildlife on their persons and through luggage. Notably, not all transporters may realize that they are participating in the illegal wildlife trade. For instance, legitimate shipping companies may be transporting wildlife products unknowingly due to the sheer number of consignments. Indeed, it has been argued that "transportation is the backbone of global trade, and traffickers in wild animals and wildlife products rely heavily on logistics, land, air and sea carriers to smuggle illicit goods" (TRAFFIC 2015).

F. Sellers

Those involved in selling illegal wildlife products vary significantly—from those working in local markets to well-established, legitimate collectors organizations. For example, an individual who works in a local market may sell captured live animals in addition to other goods (Leberatto 2016), while (illegal) caviar may only be sold by trusted sellers to extravagant restaurants (Saffron 2002).

In addition to physical markets, the expansion and use of the Internet has contributed to the development of the illegal wildlife trade (Laufer 2010). Despite increased awareness and vigilance by some online companies, the proliferation of illegal wildlife products continues on the Internet (e.g., Hastie and McCrea-Steele 2014). Moreover, it is believed that the existence of the "dark web" provides another avenue for sellers to interact with would-be buyers (Khan 2013).

G. Organized Crime, Terrorist, and Rebel Groups

The belief that organized crime syndicates are directly involved in the illegal wildlife trade is fueled by the high profits associated with specific wildlife products (e.g., ivory, rhino horn) and the ability to utilize established criminal networks and personnel, smuggling routes, and resources to entice corrupt officials (Zimmerman 2003; Elliot 2012; IFAW 2013). Media accounts further add fodder to the belief that organized crime syndicates are actively engaged in wildlife trafficking (Morelle 2014). However, the extent of organized crime groups involvment in the illegal wildlife trade is still up for debate. While some scholars have noted the existence of organized crime groups in particular stages (Wyatt 2009) or for specific species (Warchol 2004), others have found little or no evidence suggesting that organized crime syndicates are involved in the illegal

Page 15 of 41

wildlife trade (Leberratto 2016; Pires et al. 2015). Factoring in what is currently known about the role of organized crime in the illegal wildlife trade, it is sensible to state that more research is required, especially from a species- or product-specific perspective (see Moreto and Lemieux 2015a). Moreover, it is important to note that the level of organization required for a trade to operate does not equate to the overall trade being within the stronghold of organized crime groups. While organized crime may be present at certain stages of the illegal wildlife crime and organized-crime indicators may be present (see Sellar, 2007), it does not seem to be evident at *all* stages of the trade.

Finally, anecdotal evidence suggests that terrorist and rebel groups may also be involved in wildlife trafficking (Neme 2009; WWF 2012). It is believed that such groups hunt, trade, and smuggle wildlife products to fund their operations (Lawson and Vines 2014). For example, in 2013, the UN Secretary-General's report emphasized the potential role that the Lord's Resistance Army may have in elephant poaching and the illegal ivory market.¹

H. Consumers

The consumers of the illegal wildlife market are quite diverse and include those who seek wildlife products for cultural medicines and traditions, luxurious clothing, textiles, food delicacies, and exotic pets. Although much of the literature tends to focus on the transnational nature of the illegal wildlife trade, it is important to note that local individuals can also be consumers, and, in some instances, the local underground market may be significant, as is the case for illegal caviar in Russia (see http://assets.panda.org/downloads/Sturgeon factsheet.pdf).

V. Supply Markets, Transit Countries, and Demand Markets

Similar to the actors, the supply and demand markets will vary among the different wildlife products. In general, however, it is believed that both the legal and illegal wildlife market flow from emerging to developed countries (Reeve 2002; Roe et al. 2002; Duffy 2010; Rosen and Smith 2010; Lawson and Vines 2014). Open-air illicit markets are typically found in cities large and small throughout biodiverse regions, such as Latin America, Africa, and Southeast Asia, where an abundance of wildlife is for sale. Locals and indigenous people, often from impoverished communities, have historically trapped and traded (or poached and trafficked) wildlife to supply such markets and supplement their incomes (Leberatto, 2016; Pires et al. 2016; Roe, 2002; TRAFFIC, 2008). From the demand perspective, the main destination areas are believed to be China, the United States, and the European Union (Wyatt 2013). However, this will also differ based on the mode of transaction (e.g., Internet), which further highlights the challenges in

Page 16 of 41

generalizing the illegal market. Lastly, due to their geographical location, lax laws, and number of available ports of entry, several countries are considered to be transit countries. For instance, the illegal trade in both ivory and rhino horn have demonstrated several transit countries within the overall market (Milliken 2014).

VI. Types of Wildlife Trafficking and Commonly Trafficked Wildlife

Wildlife is poached and trafficked for a number of reasons, which include alternative medicines, accessories or luxury items, bushmeat, and the live pet trade. Poaching and trafficking wildlife for their supposed medicinal qualities, otherwise known as traditional Asian medicines (TAM), is one of the largest contributors to the animal parts trade both in terms of volume and species affected. For example, animal parts, such as tiger bones and genitals, bear gallbladders, deer fetuses, elephant skins and other parts, antelope horns, rhino horns, pipehorses, and snakes are trafficked for use in TAMs (Martin-Smith et al. 2003; Li and Wang 1999). Wild birds, shark fins, bear paws, and dried abalone are trafficked as a delicacy in Asia and Europe (Shepherd and Shepherd 2010; Forero 2006; Papp 2008; To et al. 2006). Other animal parts, such as skins (e.g., leopards, crocodiles) and scales (e.g., pangolins) are used as wearing apparel (Hutton and Webb 2003; Bräutigam et. al. 1994); while elephant ivory is used to make pendants, necklaces, rings, and other fashion accessories (Martin and Vigne 2011; Martin and Stiles 2004). Finally, wild animals, such as parrots, tortoises, and macaques, are traded live as pets (Broad et al. 2003; van Lavieren 2008) and are sold to private collectors as well as circuses (Burgener 2002; Speart 1993, as cited in Warchol et al. 2003). The plowshare tortoise, for example, is the most-sought animal in the world for the live pet trade in part because of its extreme rarity. One plowshare tortoise can sell upward of US\$100,000 (IUCN, 2015a; Finnegan, 2012) as a consequence of so few remaining in the world.

A great variety of species are routinely poached and trafficked, but some much more often than others. Of the studies that have examined poaching variation among wildlife, they have all found an unequal distribution of poaching. That is, some species are highly poached and trafficked while others are not (Petrossian et al. 2016; Petrossian and Clarke 2014; Pires and Clarke 2012; Nijman, 2010; Robbins 2003). The distribution of poached species appears to resemble the 80-20 rule (Clarke and Eck 2005), in that a small percentage of the world's species account for a great majority of total poaching. For example, only a few groups of wildlife, namely mammals and reptiles, account for the great share of wildlife seized entering the United States (Petrossian et al. 2016). The following section will focus on the most commonly poached animals that have garnered the most attention from conservationists and policymakers either because they have experienced a rapid population decline or are on the brink of extinction.

A. Rhinos

Once widely abundant and dispersed, rhino populations currently number only in the thousands and are only found in protected parks in a few Asian and African countries (see www.savetherhino.org). Five species of rhino remain today, three of which are critically endangered, and the remaining two are vulnerable and near threatened (IUCN 2015b). While there are several threats rhinos face, poaching is the most serious among them. Rhinos are presently one of the most hunted animals on the planet for their rhino horn, and this is predominantly occurring in South Africa, where the largest population of rhinos exist (www.savetherhino.org). The main demand markets for rhino horn are China and Vietnam (Beech and Perry 2011; TRAFFIC 1997; Wilson-Wilde 2010), where people believe rhino horn can cure a variety of ailments, which recently includes cancer (Moore 2011b; Wilson-Wilde 2010). Evidence, however, casts doubt on whether rhino horn ameliorates any ailments (Nowell 2012; Moore 2011b). In addition to using horn for TAM, wealthy individuals in Asia acquire it simply to show status (Mander 2012). Rhino horn is also used for *jambiya* dagger handles in Yemen (www.savetherhino.org).

Rhino horn cannot be commercially traded in international markets since CITES outlawed it in 1977. As a result of this and conservation efforts on the ground, the white rhino and the greater one-horned rhino experienced incredible growth during the following three decades. Until 2008, rhino poaching was relatively rare in South Africa with an average of 15 losses per year (Eloff and Lemieux 2014). In 2008, however, poaching escalated to 83 losses and has increased every year since. By 2014, 1,215 rhinos were poached in South Africa alone (www.savetherhino.org). With diminishing populations and a growing demand from Vietnam, the price of rhino horn has increased dramatically. Rhino horn was recently valued at US\$35,000-\$60,000 per kilo (Eustace 2011; Mander 2012; Moore 2011b), making the average rhino horn (2.5 kilos) (Stoddard 2012) cost between US\$87,500 and \$150,000. Rhino horn is now worth more than elephant ivory or any other animal part in the world.

Given its current price in international markets, there is a tremendous incentive to be involved in either poaching, trafficking, or selling rhino horn to consumers (Eloff and Lemieux 2014). According to Jackson (2012), there are four types of poachers: subsistence, commercial, skilled, and chemical. The least skilled are subsistence poachers, who are generally poor locals who often use snares and machetes to trap and kill rhinos (Eloff and Lemieux 2014). Commercial poachers are suspected of being the most common, and they largely account for the drastic increase in poaching in South Africa since 2008. They are more likely to use high-powered guns and work in teams of four to six individuals with military backgrounds. Once the rhino is killed, they can quickly dispose of the horn to middlemen. Skilled poachers are far more sophisticated and will often kill rhinos with one shot. They will quickly remove the horn and exit the area before rangers are aware of the loss. Finally, chemical poachers use tranquilizer darts to sedate rhinos, which sometimes emanates from a helicopter. While a highly

Page 18 of 41

sophisticated form of removing rhino horn, it is relatively rare compared to commercial poaching (Jackson 2012; www.stoprhinopoaching.com).

Apart from poaching rhinos, there are other ways that rhino horn is obtained to supply demand markets in Asia. "The very people responsible for 'protecting' rhinos are sometimes exploiting their professional networks and unrestricted access to the animals to earn large sums of money" (Eloff and Lemieux 2014: 21). This can take the form of pseudo-conservation (e.g., mismanaging rhino horn stockpiles) or pseudo-hunting²—both have become increasingly common in South Africa (Ayling 2013; Milliken and Shaw 2012; Beech and Perry 2011; Eloff and Lemieux 2014). In addition to these two methods, stealing horns from museums, safari lodges, game reserves, and taxidermies has become more commonplace (Milliken and Shaw 2012).

A number of initiatives can be implemented to reduce the likelihood of rhino poaching. A recent study found that rhino poaching is spatially concentrated within Kruger National Park in South Africa and that it occurs more often near roads and the Mozambique border (Eloff and Lemieux 2014). Eloff and Lemieux (2014) suggest a number of solutions, particularly technologically savvy ones (e.g., remote sensing, gunshot detectors, unmanned aerial systems or drones) that can alert rangers to where suspicious activity is occurring in real-time thereby *increasing the risk* of apprehension of offenders. On the prevention side, Eloff and Lemieux (2014) recommend controlling access to parks so that there only a few ways in or out of the park. Authorities can also digitally record license plate numbers of cars and visitor identification information to deter poachers from using vehicles. Lastly, routine vehicle searches at park exits can be conducted to look for rhino horn. In addition to these recommendations, more needs to be done about reducing demand for rhino horn. Public education campaigns aimed at the Chinese and Vietnamese people could theoretically reduce demand by alerting them to the extinction risk of rhinos and the lack of medicinal benefits from rhino horn ingestion.

B. Pangolins

Pangolins, also known as scaly anteaters, are mammals (Nowak and Paradiso 1983) that play an important ecological role by feeding on ants, termites, and the like. While pangolins do not receive as much media attention as overexploited photogenic species, they are among the most poached mammals in the world (Sutter 2014). About 100,000 to 135,000 pangolins per year are needed to meet the demand in China, which is the largest demand market in the world (Shibao, Guangzhi, and Qianxiang 2005). China has had a long history of using pangolin for consumption and as a traditional Chinese medicine (TCM) (Pantel and Chin 2009; Shepherd 2009; Wu et al. 2004). Pangolin scales are believed to be an important ingredient for treating a variety of ailments (Yue 2009; Acosta-Lagrada 2008; Phalla 2008; Wang 2008) and are also used to make leather accessories (Phalla 2008). As a result of this unsustainable demand, pangolins are commercially extinct in the wild in China (Wu et al. 2004). China's neighbors have also

Page 19 of 41

been depleted of their pangolin population with the supply currently coming from Indonesia and Malaysia (Semiadi, Darnaedi, and Arief 2008).

There are eight different species of pangolin worldwide, four located in Southeast Asia and four located in Africa. The two most commonly traded pangolin species in Asia are the Manis javanica and Manis pentadactyla, and both species are listed as "critically endangered" by the IUCN (2015b). The other two Asian species of pangolins—Manis culionensis and Manis crassicaudata—are considered "endangered." In Africa, all four pangolin species, Smutsia temminckii, Phataginus tricuspis, Uromanis tetradactyla, and Smutsia gigantea, are classified as "vulnerable." The African species' decline is attributed to bushmeat exploitation and the exportation of pangolin meat and scales used for TCM in Asia (IUCN 2015b). While the African pangolin species are commonly poached, they may be less endangered than their counterpart Asian species for a number of reasons. Trafficking pangolins from Africa to China is not easy, especially if they are alive. There is some difficulty in trafficking live pangolins far distances as they are very hard to keep in captivity and are easily detected. Live pangolins often die in transit to their destination market due to disease and unsanitary conditions (Clark, Van Thai, and Phuong 2008). Additionally, the Chinese demand is generally supplied with pangolins and their byproducts from neighboring countries because the distance between supply and demand markets is much closer and thereby easier.

The price increase of pangolins over time is directly related to their declining population in the wild, making it difficult to disincentivize hunters from poaching. A study conducted in China revealed that prices are 100 times higher than those in the early 1980s and 20 times higher than in the 1990s (Challender et al. 2015a; Shibao et al. 2005). A kilogram of pangolin scales will now sell for around US\$199 (Guilford 2014). Served as meat, one pangolin can sell for upward of US\$1,000 (Allgood 2013), which can demonstrate a person's socioeconomic status.

Pangolins are exceptionally vulnerable to overexploitation because they have a very slow reproduction rate, do not breed easily in captivity (Lim and Ng 2007; Shepard 2009), and are easily hunted. Pangolins main defense against predators is to roll up in a ball, which is highly effective when encroached by a nonhuman predator, but highly ineffective against humans (Lekagul and McNeely 1977; Pantel and Awang Anak 2010). Once a hunter finds a pangolin, no real effort or skill is required to capture them. Various poaching methods are used to capture pangolins and are contingent on the country's landscape and the species being hunted, as some species are arboreal and others are terrestrial. Hunters find pangolins through six common techniques: trained dogs, tracking strategies, pangolin-specific traps, nonselective traps, spotlights, and opportunistic encounters (Newton et al. 2008). Opportunistic encounters come in the form of hunters stumbling upon the opportunity to capture a pangolin while hunting other animals or cutting down a tree for firewood (Sterling, Hurley and Le 2006).

All Southeast Asian countries already have laws in place that make hunting pangolins illegal. Additionally, all Asian species are listed under Appendix II of CITES and have a zero trade quota (Challender et al. 2015b; Kang and Phipps 2003). Antipoaching laws and international treaties have done little to thwart the trade in pangolins—most species are in danger of becoming extinct. The incentive to poach and traffic pangolins will be difficult to eradicate or reduce as the ever increasing price per kilo only further incentivizes poaching. A number of solutions that target both the demand- and supplysides of the equation can theoretically reduce the trade. Reducing demand can come in the form of public education campaigns in China and other Southeast Asian countries where demand is the greatest. Also, medicinal alternatives to pangolin scales can be provided in the form of dry seeds of cowherb, known as Wang Bu Liu Xing, which has the same effectiveness (Wang 2008). On the supply side, law enforcement can make it more difficult to peach and traffic pangelins by targeting hot areas (i.e., forests, roads, borders, ports) where pangolin poaching and trafficking disproportionately occur. One study found that setting up road stops on a major road used for trafficking pangolins and other wildlife led to a considerable seizure amount (Lee et al. 2005). Unfortunately, research on hot spots related to the illicit pangolin trade has been largely ignored, and this is where future research could aid law enforcement strategies in thwarting the trade.

C. Parrots

Unlike other commonly poached animals, parrots are among the few that are poached live for the pet trade. Wild parrots are found throughout most regions of the world but are more typical in areas with warmer climates. While their presence is vast, parrot poaching disproportionately occurs in Central and South America, otherwise known as the neotropics. Historically, parrots have been taken from the wild in the neotropics laws permitting or not. Parrot poaching was largely committed for personal ownership or for local sale (Cantu et al. 2007). Such practices did not push many species to the brink of extinction, because it was limited in scope and parrots were far more abundant in prior decades and centuries. It was not until the 1970s that poaching became an industrialized enterprise in large part to the demand from the United States and European nations (Pires and Clarke 2012; Cantu et al. 2007). Once the illegal parrot trade became a global phenomenon, coupled with deforestation, parrot populations began to rapidly decline for a great many species (Clarke and Rolf 2013; Juniper and Parr 1998; Howell and Webb 1995), making parrots one of the most-endangered group of birds on the planet (Cockle et al. 2007; Pain et al. 2006; Wright et al. 2001).

By the end of the 20th century, CITES banned the trade of most parrot species and the United States banned the importation of wild parrots (Armstrong et al. 2001). Ostensibly, this resulted in a large decline in the illegal parrot trade at the international level (Cooney and Jepson 2006; Wright et al. 2001). Recent studies, however, have shown that the domestic trade in parrots is a far greater problem than the international trade. A Mexican report based on interviews with trappers (Cantu et al. 2007) found upward of 78,500 parrots were being poached annually and the vast majority were for domestic demand. Three separate market surveys in Bolivia (Herrera and Hennessey 2007), in Peru (Gastanaga et al. 2011), and in Brazil (Regueira and Bernard 2012) found similar results in terms of the volume and the domestic nature of the trade. Such evidence suggests that demand for pet ownership of parrots within neotropical countries is very high (Weston and Memon 2009; Cantu et al. 2007; Gonzalez 2003), and current poaching levels are unsustainable for the great majority of species in these countries (Gastanaga et al. 2011; Pires 2012; Cantu et al. 2007).

While a majority of species are affected by the illegal parrot trade, some are more affected than others. Recent criminological research has applied the theft variation model, CRAVED (Clarke 1999), to understand why poaching variation exists among the more than 330 parrot species in the world. The conventional wisdom in the field of conservation was that parrot species that are disproportionately poached would be the most attractive, valuable, and/or rare. Such thinking insinuates that poachers will target a particular type of species based on profitability and/or their perception of consumer demand (Pires and Clarke 2012). While offenders might seek such species in the wild, they may not be as available or accessible as other species (Pires 2012). In fact, several criminological studies have found this to be the case. Species that are less valuable, less rare, and that are average appearing in terms of "attractiveness" are the ones that are

Page 22 of 41

poached and trafficked most frequently (Pires 2015a; Pires and Petrossian 2016; Pires and Clarke 2012, 2011). Factors of opportunity actually explain poaching variation the best. That is, species that are closer to illicit markets, have larger ranges, are more abundant, nest closer to the ground, and are more concealable are significantly related to higher poaching counts in several countries (Pires 2015a, 2015b; Pires and Petrossian 2016; Pires and Clarke 2012, 2011). However, one study has found that the more attractive species in Mexico were poached more often even while controlling for opportunistic factors (Tella and Hiraldo 2014).

More recent criminological research has been able to map where and when poaching is occurring in Bolivia. Pires et al. (2015b) found that parrot poaching is highly concentrated in space, in time, on particular species, and on particular methods of obtaining them. Just seven municipalities, or 11 percent of the total area in a department of Bolivia (the political equivalent of a US state), accounted for over 80 percent of poaching. Further, about 50 percent of poaching occurred in a three-month window during the summer. Overall, the implications of such findings are to allocate conservation and law enforcement resources to particular areas, at particular times, and protect the mostendangered species along with the most poached ones. In doing so, the largest reduction in the illegal parrot trade can be achieved while using limited resources more prudently.

VII. Implications for Future Research

As this systematic review of the literature has shown, the illegal wildlife trade varies greatly in nature and in scope. There are some common patterns among species in the trade in relation to methods used, actors involved, demand and supply countries, and purposes of the trade; but the trade also operates uniquely for each species. Acknowledging that the trade operates in slightly different ways for each species is important to consider as it relates to interventions to reduce the trade. Policymakers cannot simply reduce the trade in all species using the same tool. What works with one species may not work with another because there are numerous contextual factors involved. Policymakers need to use a myriad of tools that are tailored toward a specific species and the larger problem at hand.

Most research on the topic emanates from the conservation sciences as they regularly obtain grants, conduct field research, and, even at times, implement solutions to reduce wildlife crime. Evaluations of such interventions are lacking and need to be conducted on a frequent basis going forward so that both the academic and policy communities can learn what works with respect to how, why, and with which species. Criminologists and criminal justice scholars have a role to play in these evaluations since they are best suited to understand why crime occurs and how it may be prevented through a number of ways. Hopefully, interdisciplinary collaborations, such as those between conservationists and

Page 23 of 41

criminologists, can foster a better understanding of how to mitigate the illegal wildlife trade and conserve at-risk species.

References

Abensperg-Traun, M. (2009). CITES, sustainable use of wild species and incentive-driven conservation in developing countries, with an emphasis on southern Africa. *Biol. Conserv.* 142, 948–963.

Acosta-Lagrada, L. (2008). Conservation and management of the Malayan pangolin in Palawan Providence, Philippines. TRAFFIC Southeast Asia. Proceedings of the workshop on trade and conservation of pangolins Native to South and Southeast Africa, 29–39.

Agnew, D., Pearce, J., Peatman, T., Pitcher, T. J., and Pramod, G. (2008). The global extent of illegal fishing. Fisheries Ecosystems Restoration Research Fisheries Centre, University of British Colombia and Marine Resources Assessment Group, Vancouver and London.

Ahmed, A. (2010). Imperiled custodians of the night: A study on the illegal trade, trapping and use of owls in India. TRAFFIC India. http://www.traffic.org/species-reports/traffic_species_birds12.pdf.

Allgood, B. (2013, June 6). Criminal nature—the dangerous links between poaching and organized crime. International Fund for Animal Welfare. Retrieved Oct 1, 2015, http://www.ifaw.org/united-states/news/criminal-nature-%E2%80%93-dangerous-links-between-poaching-and-organized-crime.

Andresen, M. A. (2014). *Environmental criminology: Evolution, theory, and practice*. New York: Routledge.

Armstrong, M. C., Farinato, R. H., and Telecky, T. M. (2001). A wing and a prayer: Birds and their protection under law. *Journal of Avian Medicine and Surgery*, 15(4), 310–315.

Ayling, J. (2013). What sustains wildlife crime? Rhino horn trading and the resilience of criminal networks. *Journal of International Wildlife Law and Policy*, 16(1), 57–80.

Barnosky, A. D., Matzke, N., Tomiya, S., Wogan, G. O., Swartz, B., Quental, T. B., and Mersey, B. (2011). Has the Earth/'s sixth mass extinction already arrived?. *Nature*, *471*(7336), 51-57.

Beech, H., and Perry, A. (2011). Killing fields: How Asia's growing appetite for traditional medicine is threatening Africa's rhinos. *Time*, 177 (24), 40–47.

Beirne, P. (1999). For a Nonspeciesist Criminology: Animal Abuse as an Object of Study. *Criminology*, 37: 117–148.

Bell, S., Hampshire, K. and Topalidou, S. (2007). The political culture of poaching: A case study from northern Greece. *Biodiversity and Conservation*, 16: 399-418.

Page 24 of 41

Bowman, M. (2013). A tale of two CITES: Divergent perspectives upon the effectiveness of the wildlife trade convention. *Review of European, Comparative and International Environmental Law*, 22(3), 228–238.

Bräutigam A., Howes, J., Humphreys, T., Hutton, J. (1994). Recent Information on the Status and Utilization of African Pangolins. *TRAFFIC Bulletin*, 15(1), 15–22.

Briggs, D., Courchamp, F., Martin, R., Possingham, H. P. (2013). Legal trade of Africa's rhino horns. *Science*, 339, 1038–1039.

Brisman, A. (2015). "Multicolored" green criminology and climate change's achromatopsia. *Contemporary Justice Review*, 18, 178–196.

Brisman, A. and South, N. (2013). A green-cultural criminology: An exploratory outline. *Crime Media Culture: An International Journal*, *9*, 115–135.

Broad, S., Mulliken, T., Roe, D. (2003). The nature and extent of legal and illegal trade in wildlife. In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 3–22. London: Earthscan.

Burgener, M. (2002). The Law and Policy Working Group. Endangered Wildlife, 39, 69.

Butchart, S., et al. (2010). Global biodiversity: Indicators of recent declines. *Science 328*, 1164–1168.

Cantu, J. C. G., Saldana, M. E. S., Grosselet, M., and Gamez, J. S. (2007). *The illegal parrot trade in Mexico: A comprehensive assessment*. Mexico and Washington, DC: Defenders of Wildlife.

Ceballos, G., Ehrlich, P. R., Barnosky, A. D., García, A., Pringle, R. M., and Palmer, T. M. (2015). Accelerated modern human-induced species losses: Entering the sixth mass extinction. *Science Advances*, 1(5), e1400253.

Challender, D. W., Harrop, S. R., and MacMillan, D. C. (2015a). Understanding markets to conserve trade-threatened species in CITES. *Biological Conservation*, 187, 249–259.

Challender, D. W., Harrop, S. R., and MacMillan, D. C. (2015b). Towards informed and multi-faceted wildlife trade interventions. *Global Ecology and Conservation*, *3*, 129–148.

CITES (2015a). The CITES Appendices. Retrieved Dec. 1, 2015, https://www.cites.org/eng/app/index.php.

CITES (2015b). How CITES works. Retrieved Dec. 1, 2015, https://www.cites.org/eng/disc/how.php.

CITES (2015c). The CITES species. Retrieved Dec. 1, 2015, https://www.cites.org/eng/disc/species.php.

Page 25 of 41

Clark, L., Van Thai, N., and Phuong, T. Q. (2008). A long way from home: The health status of Asian pangolins confiscated from the illegal wildlife trade in Vietnam. TRAFFIC Southeast Asia. Proceedings of the workshop on trade and conservation of pangolins native to South and Southeast Africa, 111–118.

Clarke, R. V. (1980). "Situational" crime prevention: Theory and practice. *British Journal of Criminology*, 20, 136–147.

Clarke, R. V., ed. (1997). *Situational crime prevention: Successful case studies*, 2nd ed. Monsey, NY: Criminal Justice Press.

Clarke, R. V. (1999). Hot products: Understanding, anticipating and reducing demand for stolen goods. Police Research Series, Paper 112: Home Office.

Clarke, R. V. (2004). Technology, criminology, and crime science. *European Journal of Criminal Policy and Research* 10, 55-63.

Clarke, R. V. (2009). Situational crime prevention: Theoretical background and current practice. In *Handbook on Crime and Deviance* ed. M. D. Krohn, A. J. Lizotte. and G. P. Hall, 259–276. New York: Springer.

Clarke, R. V., and Rolf, A. (2013). Poaching, habitat loss and the decline of neotropical parrots: A comparative spatial analysis. *Journal of Experimental Criminology*, 9(3), 333–353.

Cockle, K., Capuzzi, G., Bodrati, A., Clay, R., Del Castillo, H., Velázquez, M., ... and Farina, R. (2007). Distribution, abundance, and conservation of Vinaceous Amazons (Amazona vinacea) in Argentina and Paraguay. *Journal of Field Ornithology*, 78(1), 21–39.

Cohen, L. E., & Felson, M. (1979). Social change and crime rate trends: a routine activity approach. *American Sociological Review*, 44(4), 588-608.

Cole, R. (2012). The effect of international trade bans on the population of endangered species. *Penn State Journal of International Affairs*, 2(1), 35–53.

Conrad, K. (2012). Trade bans: A perfect storm for poaching. *Tropical Conservation Science*, 5(3), 245–254.

Cooney, R., and Jepson, P. (2006). The international wild bird trade: What's wrong with blanket bans?. *Oryx*, 40(1), 18–23.

Cornish, D. B. (1994). The procedural analysis of offending and its relevance for situational prevention. In *Crime prevention studies*, vol. 3, ed. R. V. Clarke, 151–196. Monsey, NY: Criminal Justice Press.

Cornish, D., and Clarke, R. (1986). The reasoning criminal. New York: Springer.

Courchamp, F., Angulo, E., Rivalan, P., Hall, R. J., Signoret, L., Bull, L., and Meinard, Y. (2006). Rarity value and species extinction: the anthropogenic allee effect. *Plos One: Biology*, 4(12), e415.

Couzens, E. (2013). CITES at forty: Never too late to make lifestyle changes. *Review of European, Comparative and International Environmental Law, 22*(3), 311–323.

Daut, E. F., Brightsmith, D. J., Mendoza, A. P., Puhakka, L., and Peterson, M. J. (2015). Illegal domestic bird trade and the role of export quotas in Peru. *Journal for Nature Conservation*, *27*, 44–53.

Dickson, B. (2003). What is the goal of regulating wildlife trade? Is regulation a good way to achieve this goal? In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 23–32. London: Earthscan.

Duffy, R. (2010). *Nature crime: How we're getting conservation wrong*. New Haven, CT: Yale University Press.

du Plessis, M. A. (2000). CITES and the causes of extinction. In *Endangered species* threatened convention: The past, present and future of CITES, ed. J. Hutton, B. Dickson, 13–25. London: Earthscan.

Clarke, R. V., and Eck, J. E. (2005). *Crime analysis for problem solvers*. Washington, DC: Center for Problem Oriented Policing.

Eliason, S. L. (1999). The illegal taking of wildlife: Toward a theoretical understanding of poaching. *Human Dimensions of Wildlife*, 4, 27–39.

Eliason, S. L. (2003). Throwing the book versus cutting some slack: Factors influencing the use of discretion by game wardens in Kentucky. *Deviant Behavior*, *24*, 129–152.

Eliason, S. L. (2006). Factors influencing job satisfaction among state conservation officers. *Policing: An International Journal of Police Strategies and Management*, 29, 6–18.

Eliason, S. L. (2008). Wildlife crime: Conservation officers' perceptions of elusive poachers. *Deviant Behavior*, 29: 111–128.

Eliason, S. L. (2012). Trophy poaching: A routine activities perspective. *Deviant Behavior*, 33, 72–87.

Eliason, S. L. and Dodder, R. A. (1999). Techniques of neutralization used by deer poachers in the western U.S.: A research note. *Deviant Behavior*, 20, 233–252.

Elliot, L. (2012). Fighting transnational environmental crime. *Journal of International Affairs*, 66: 87–104.

Eloff, C. and Lemieux, A. M. (2014). Rhino poaching in Kruger National Park, South Africa. In *Situational Prevention of Poaching*, ed. A. M. Lemieux, 18–43. New York, New York: Routledge.

Environmental Resources Management (ERM) (1996). Study on how to improve the effectiveness of CITES. Final Report to the Standing Committee of CITES, Lausanne.

Eustace, M. (2011). Current supply and demand in the rhino horn market—A model for regulating the rhino horn trade (rewards to trade). In *Perspectives on Dehorning and Legalized Trade in Rhino Horn as Tools to Combat Rhino Poaching*, ed. B. G. A. Daly, et al. Proceedings of a workshop assessing legal trade in rhino horn as a tool in combating poaching as well as a detailed assessment of the efficacy of dehorning as a deterrent to poaching, 10. Johannesburg, South Africa: Endangered Wildlife Trust.

Felson, M., and Poulsen, E. (2003). Simple indicators of crime by time of day. *International Journal of Forecasting*, 19(4), 595–601.

Fiadjoe, Y. (2004). Cities in Africa: An examination of domestic implementation and compliance. *Sustainable Development and Policy*, 4, 38–46.

Finnegan, W. (2012, Jan 23). Slow and steady: A Manhattan night-life baron's race to save an ancient species. *The New Yorker*.

Franke, J., and Telecky, T. M. (2001). *Reptiles as pets: An examination of the trade in live reptiles in the United States*. Humane Society of the United States.

Frisina, M. R., and Tareen, S. N. A. (2009). Exploitation prevents extinction: Case study of endangered Himalayan sheep and goats. *Recreational Hunting, Conservation and Rural Livelihoods: Science and Practice*, 141–156.

Forero, J. (2006, Jan 5). Hidden Cost of Shark Fin Soup: Its Source May Vanish. *The New York Times, International*. Retrieved from: http://www.nytimes.com/2006/01/05/international/americas/05sharks.html? r=0.

Forsyth, C. J. (1993). Chasing and catching "bad guys": The game warden's prey. *Deviant Behavior*, 14, 209-226.

Forsyth, C. J., Gramling, R. and Wooddell, G. (1998). The game of poaching: Folk crimes in Southwest Louisiana. *Society and Natural Resources*, 11, 25–38.

Fuchs, C. (2010). Convention in international trade in endangered species of wild and flora (CITES)—conservation efforts undermine the legality principle. In: von Bogdandy, A., et al. (Eds.), The Exercise of Public Authority by International Institutions. In: Beiträge zum Auslandischen Öffentlichen Recht und Völkerrecht, vol. 210.

Gastanaga, M., MacLeod, R., Hennessey, B., Nunez, J. U., Puse, E., Arrascue, A., and Engblom, G. (2011). A study of the parrot trade in Peru and the potential importance of internal trade for threatened species. *Bird Conservation International*, 21(1), 76–85.

Geist, H. J., and Lambin, E. F. (2002). Proximate causes and underlying driving forces of tropical deforestation tropical forests are disappearing as the result of many pressures, both local and regional, acting in various combinations in different geographical locations. *BioScience*, *52*(2), 143–150.

Gibbs, C., Gore, M. L., McGarrell, E. D., and Rivers, L, III. (2010). Introducing conservation criminology: Towards interdisciplinary scholarship on environmental crimes and risks. *The British Journal of Criminology*, *50*, 124–144.

Gonzalez, J. A. (2003). Harvesting, local trade, and conservation of parrots in the Northeastern Peruvian Amazon. *Biological Conservation*, 114(3), 437–446.

Gore, M. L. (2011). The science of conservation crimes. *Conservation Biology*, 25, 659-661.

Guilford, G. (2014, Jan 27). Demand for traditional Chinese medicine is killing off the world's quirkiest animal. *Quartz*. http://qz.com/170554/demand-for-traditional-chinese-medicine-is-killing-off-the-worlds-quirkiest-animal/.

Hall, R. J., Milner-Gulland, E. J., and Courchamp, F. (2008). Endangering the endangered: The effects of perceived rarity on species exploitation. *Conservation Letters*, 1(2), 75–81.

Halsey, M. (2004). Against "green" criminology. *The British Journal of Criminology*, 44, 833–853.

Haken, J. (2011). Transnational crime in the developing world. Global Financial Integrity.

Han, C. (2014). A survey of customs administration perceptions on illegal wildlife trade. WCO Research Paper, No. 34.

Hastie, J. and McCrea-Steele, T. (2014). Wanted—dead or alive: Exposing online wildlife trade. London: IFAW.

Herbig, F. J. W. (2011). South African conservation crime and routine activities theory: A causal nexus. Inaugural lecture, July 20. University of South Africa, Pretoria.

Herbig, F. J. W. (2014). The phronesis of conservation criminology phraseology: a genealogical and dialectical narrative. Phronimon, 15(2), 1–17.

Herrera, M., and Hennessey, B. (2007). Quantifying the illegal parrot trade in Santa Cruz de la Sierra, Bolivia, with emphasis on threatened species. *Bird Conservation International*, 17(4), 295–300.

Howell, S. N., and Webb, S. (1995). *A guide to the birds of Mexico and northern Central America*. Oxford University Press.

Hutton, J., and Webb, G. (2003). Crocodiles: Legal trade snaps back. *The Trade in Wildlife: Regulation For Conservation*, 108, 114.

Huxley, C. (2000). CITES: The vision. In *Endangered species, threatened convention: The past, present and future of CITES*, ed., J. Hutton and B. Dickson, 3–12. London: Earthscan.

IFAW. (2013). Criminal nature: The global security implications of the illegal wildlife trade. Washington, DC: IFAW.

Interpol (2015). Wildlife crime. Retrieved Dec 1, 2015, http://www.interpol.int/public/EnvironmentalCrime/Wildlife/Default.asp.

IUCN (2015a). Assessment of the Astrochelys Yniphora. Retrieved Dec 2, 2015, http://www.iucnredlist.org/details/9016/0.

IUCN (2015b). IUCN red list of threatened species. Retrieved Dec 2, 2015, http://www.iucnredlist.org/search.

Jachmann, H. (2003). Elephant poaching and resource allocation for law enforcement. In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 100–107. London: Earthscan.

Jackson, T. (2012). The crisis. Africa Geographic, 20 (3), 30-36.

Jacobson, S. K., and Duff, M. D. (1998). Training idiot savants: The lack of human dimensions in conservation biology. *Conservation Biology*, 12, 263–267.

Juniper, T., and Parr, M. (1998). Parrots: A guide to the parrots of the world. Pica, Sussex.

Kang, S., and Phipps, M. J. (2003). A question of attitude: South Korea's traditional medicine practitioners and wildlife conservation. TRAFFIC East Asia.

Kahler, J. S. and Gore, M. L. (2012). Beyond the cooking pot and pocket book: Factors influencing noncompliance with wildlife poaching rules. *International Journal of Comparative and Applied Criminal Justice*, *36*, 103–120.

Kahler, J. S., Roloff, G. J. and Gore, M. L. (2013). Poaching risks in community-based natural resource management. *Conservation Biology*, *27*, 177–186.

Khan, D. (2013, May 28). Slipping through the net: The illegal online wildlife trade. Retrieved Dec 22, http://www.huffingtonpost.co.uk/vidhi-doshi/illegal-online-wildlife-trade b 3337537.html.

Kievit, H. (2000). Conservation of the Niles crocodile: Has CITES helped or hindered? In *Endangered species threatened convention: The past, present and future of CITES*, ed. J. Hutton, B. Dickson, 88–97. London: Earthscan.

Larriera, A., Siroski, P., and Imhof, A. (2010). Crocodilians ranching program in Argentina: Twenty years of success. Crocodiles: Proceedings of the 20th Working Meeting of the Crocodile Specialist Group. IUCN, Gland, Switzerland, and Cambridge, UK, 73–76.

Laufer, P. (2010). Forbidden creatures: Inside the world of animal smuggling and exotic pets. Guildford, CT: Lyons Press.

Lavorgna, A. (2014). Wildlife trafficking in the internet age. Crime Science, 3, 5.

Lawson, K., and Vines, A. (2014). *Global impacts of the illegal wildlife trade: The costs of crime, insecurity, and institutional erosion*. London: The Royal Institute of International Affairs.

Laycock, G. (2005). Defining crime science. In *Crime Science: New approaches to preventing and detecting crime*, ed. M. J. Smith and N. Tilley, 3–34. Portland, OR: Willan.

Leader-Williams, N. (2003). Regulation and protection: successes and failures in rhinoceros conservation. In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 89–99. London: Earthscan.

Leberatto, A. C. (2016). Understanding the illegal trade in live wildlife species in Peru. *Trends in Organized Crime*. doi: 10.1007/s12117-015-9262-z.

Lee, R. J., Gorog, A. J., Dwiyahreni, A., Siwu, S., Riley, J., Alexander, H., and Ramono, W. (2005). Wildlife trade and implications for law enforcement in Indonesia: A case study from North Sulawesi. *Biological Conservation*, 123(4), 477–488.

Lekagul, B., and McNeely, J. A. (1977). Mammals of Thailand. Mammals of Thailand.

Lemieux, A. M. ed. (2014). Situational Prevention of Poaching. New York: Routledge.

Lemieux, A. M., and Clarke, R. V. (2009). The international ban on ivory sales and its effects on elephant poaching in Africa. *British Journal of Criminology*, 49(4), 451–471.

Li, W. and Wang, H. (1999). Wildlife Trade in Yunnan Province, China, at the Border with Vietnam. *TRAFFIC Bulletin*, 18(1), 21–30.

Lichtenstein, G. (2009). Vicuña conservation and poverty alleviation? Andean communities and international fibre markets. *International Journal of the Commons*, 4(1), 100–121.

Lim, N. T., and Ng, P. K. (2007). Home range, activity cycle and natal den usage of a female Sunda pangolin Manis javanica (Mammalia: Pholidota) in Singapore. *Endangered Species Research*, *3*, 1–8.

Page 31 of 41

Lynch, M. J., Long, M. bA., Barrett, K. L and Stretesky, P. B. (2013). Is it a crime to produce ecological disorganization? Why green criminology and political economy matter in the analysis of global ecological harms. *The British Journal of Criminology* 55, 997–1016.

Lynch, M. J. and Stretesky, P. B. (2003). The meaning of green: Constrasting criminological perspectives. *Theoretical Criminology*, 7, 217–238.

Lynch, M. J. and Stretsky, P. B. (2014). *Exploring green criminology: Toward a green criminological revolution*. Burlington: Ashgate.

McAllister, R. R., McNeill, D., and Gordon, I. J. (2009). Legalizing markets and the consequences for poaching of wildlife species: The vicuña as a case study. *Journal of Environmental Management*, 90(1), 120–130.

McClellan, K., Charles, A., and Wilson, L. (2008). Coral degradation through destructive fishing practices. *The Encyclopedia of Earth*. http://www.eoearth.org/article/Coral_degradation_through_destructive_fishing_practices.

McMullan, J. L. and Perrier, D. C. (2002). Lobster poaching and the ironies of law enforcement. *Law and Society Review*, *36*: 679–718.

MacMillan, D. C., and Han, J. (2011). Cetacean by-catch in the Korean Peninsula—by chance or by design?. *Human Ecology*, *39*(6), 757–768.

Mander, D. (2012). Damned if you do and damned if you don't—legalizing the rhino horn trade: My journey to Vietnam. Victoria: Australia: International Anti-Poaching Foundation.

Martin, R. B., 2000. When CITES works and when it does not. In *Endangered species threatened convention: The past, present and future of CITES*, ed. J. Hutton and B. Dickson, 29–37. London: Earthscan.

Martin, E. and Vigne, L. (2011). Illegal Ivory Sales in Egypt. *TRAFFIC Bulletin*, 23(1), 17–122.

Martin, E. and Stiles, D. (2004). The Ivory Markets of East Asia. *TRAFFIC Bulletin*, 20(1), 9-12.

M. E. A. (2005). *Ecosystems and human well-being scenarios: Findings of the Scenarios Working Group.* vol. 2. Washington: Island Press.

Milliken, T. (2014). *Illegal trade in ivory and rhino horn: An assessment to improve law enforcement under the wildlife traps project*. USAID and TRAFFIC.

Milliken, T., and Shaw, J. (2012). The South Africa-Viet Nam rhino horn trade nexus: A deadly combination of institutional lapses, corrupt wildlife industry professionals, and Asian crime syndicates. TRAFFIC.

Page 32 of 41

Mitchell, R. B. (2003). International environmental agreements: A survey of their features, formation, and effects. *Annual Review of Environment and Resources*, 28(1), 429-461.

Martin-Smith, K., Lam, T., and Lee, S. (2003). Trade in Pipehorses solegnathus spp. for Traditional Medicine in Hong Kong. *TRAFFIC Bulletin*, 19(3), 139–148.

Moore, A. (2011b). "Drivers of the trade in rhino horn." In *Perspectives on dehorning and legalized trade in rhino horn as tools to combat rhino poaching*, ed. B. G. Daly, et al. Proceedings of a workshop assessing legal trade in rhino horn as a tool in combating poaching as well as a detailed assessment of the efficacy of dehorning as a deterrent to poaching, 8. Johannesburg, South Africa: Endangered Wildlife Trust.

Moore, L. (2011a). The neoliberal elephant: Exploring the impacts of the trade ban in ivory on the commodification and neoliberalisation of elephants. *Geoforum*, 42(1), 51-60.

Morelle, B. (2014, Feb 12). Organised crime sets sights on wildlife. *BBC World Service*. Retrieved Apr 12, 2015, http://www.bbc.com/news/science-environment-26153516.

Moreto, W. D. (2015). Introducing intelligence-led conservation: bridging crime and conservation science. *Crime Science*, *4*, 15. doi: 10.1186/s40163-015-0030-9.

Moreto, W. D., Brunson, R. K., & Braga, A. A. (2015). "Such misconducts don't make a good ranger": Examining law enforcement ranger wrongdoing in Uganda. *British Journal of Criminology*, 55(2), 359–380.

Moreto, W. D., Brunson, R. K., & Braga, A. A. (Forthcoming). "Anything we do, we have to include the communities": Law enforcement rangers' attitudes towards and experiences of community-ranger relations in wildlife protected areas in Uganda. *British Journal of Criminology*. doi: 10.1093/bjc/azw032.

Moreto, W. D., and Clarke, R. V. (2013). Script analysis of the transnational illegal market in endangered species: Dream and reality. In *Cognition and crime: Offender decision-making and script analysis*, ed. B. Lecrec and R. Wortley, 209–220. London: Routledge.

Moreto, W. D., and Lemieux, A. M. (2015a). From CRAVED to CAPTURED: Introducing a product-based framework to examine illegal wildlife markets. *European Journal on Criminal Policy and Research*, 21, 303–320.

Moreto, W. D., and Lemieux, A. M. (2015b). Poaching in Uganda: Perspectives of law enforcement rangers. *Deviant Behavior*, *36*, 853–873.

Moyle, B. (2003). Regulation, conservation, and incentives. In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 41–51. London: Earthscan.

Muth, R. M. and Bowe, J. F., Jr. (1998). Illegal harvest of renewable natural resources in North America: Toward a typology of the motivations for poaching. *Society and Natural Resources: An International Journal*, 11, 9–24.

Neme, L. A. (2009). *Animal investigators*. New York: Scribner.

Newton, P., Van Thai, N., Roberton, S., and Bell, D. (2008). Pangolins in peril: Using local hunters' knowledge to conserve elusive species in Vietnam. *Endangered Species Research*, *6*(1), 41–53.

Ngoc, A. C. and Wyatt, T. (2013). A green criminological exploration of illegal wildlife trade in Vietnam. *Asian Journal of Criminology*, 8, 129–142.

Nijman, V. (2010). An overview of international wildlife trade from Southeast Asia. *Biodiversity and Conservation*, 19(4), 1101–1114.

Nowak, R. M., and Paradiso, J. L. (1983). Walker's mammals of the world. Vol. 2. Baltimore, MD: Johns Hopkins University Press.

Nowell, K. (2012). Assessment of rhino horn as a traditional medicine. TRAFFIC report to CITES Secretariat pursuant to contract CITES Project No. S-389, Geneva, Switzerland.

Omondi, P., Bitok, E., and Kagiri, J. (2004). Managing human-elephant conflicts: The Kenyan experience. *Pachyderm*, *36*, 80–86.

Pain, D. J., Martins, T. L. F., Boussekey, M., Diaz, S. H., Downs, C. T., Ekstrom, J. M. M., ... and Rouys, S. (2006). Impact of protection on nest take and nesting success of parrots in Africa, Asia and Australasia. *Animal Conservation*, *9*(3), 322–330.

Pantel, S. and Anak, N. A. (2010). A preliminary assessment of sunda pangolin trade in Sudah. TRAFFIC Southeast Asia Report.

Pantel, S. and Chin, S. Y. (2009). Proceedings of the workshop on trade and conservation of pangolins native to South and Southeast Asia. TRAFFIC Southeast Asia, Petaling Jaya.

Papp, D. (2008). The illegal trade in Wild Birds from South-East and Central Europe for Food in the EU. *TRAFFIC Bulletin*, 22(1), 10–12.

Parsons, E. C. M., Rose, N. A., and Telecky, T. M. (2010). The trade in live Indo-Pacific bottlenose dolphins from Solomon Islands—A CITES decision implementation case study. *Marine Policy*, *34*(3), 384–388.

Patten, R. (2010). Policing in the wild: The game wardens' perspective. *Policing: An International Journal of Police Strategies and Management*, 33, 132–151.

Petrossian, G. A. (2015). Preventing illegal, unreported and unregulated (IUU) fishing: A situational approach. *Biological Conservation*, 189, 39–48.

Page 34 of 41

Petrossian, G. A., and Clarke, R. V. (2014). Explaining and controlling illegal commercial fishing an application of the CRAVED theft model. *British Journal of Criminology*, 54(1), 73–90.

Petrossian, G., Pires, S. F., and van Uhm, D. P. (2016). An overview of seized illegal wildlife entering the United States. *Global Crime*, online first, doi: 10.1080/17440572.2016.1152548.

Phalla, S. (2008). Conservation and management of pangolins in Cambodia. TRAFFIC Southeast Asia. Proceedings of the workshop on trade and conservation of pangolins native to South and Southeast Africa, 44–48.

Phelps, J., Webb, E. L., Bickford, D., Nijman, V., and Sodhi, N. S. (2010). Boosting CITES. *Science*, 330(6012), 1752–1753.

Pimm, S. L., and Raven, P. (2000). Biodiversity: Extinction by numbers. *Nature*, 403(6772), 843–845.

Pires, S. F. (2012). The illegal parrot trade: A literature review. *Global Crime*, 13(3), 176–190.

Pires, S. F. (2015a). A CRAVED analysis of multiple illicit parrot markets in Peru and Bolivia. *European Journal on Criminal Policy and Research*, 21(3), 321–336.

Pires, S. F. (2015b). The heterogeneity of illicit parrot markets: An analysis of seven neotropical open-air markets. *European Journal on Criminal Policy and Research*, 21(1), 151–166.

Pires, S. F., and Clarke, R. V. (2011). Sequential foraging, itinerant fences and parrot poaching in Bolivia. *British Journal of Criminology*, *51*(2), 314–335.

Pires, S., and Clarke, R. V. (2012). Are parrots CRAVED? An analysis of parrot poaching in Mexico. *Journal of Research in Crime and Delinquency*, 49(1), 122–146.

Pires, S. F., and Moreto, W. D. (2011). Preventing wildlife crimes: Solutions that can overcome the "Tragedy of the Commons." *European Journal on Criminal Policy and Research*, 17(2), 101–123.

Pires, S. F., & Petrossian, G. A. (2016). Understanding parrot trafficking between illicit markets in Bolivia: an application of the CRAVED model. *International Journal of Comparative and Applied Criminal Justice*, 40(1), 63–77.

Pires, S. F., Schneider, J. L. and Herrera, M. (2016). Organized crime or crime that is organized? The parrot trade in the neotropics. *Trends in Organized Crime*, 19(1), 4–20.

Pires, S. F., Schneider J., M. Herrera, and J. Tella. (2015). Spatial, temporal and age sources of variation in parrot poaching in Bolivia. *Bird Conservation International*, 1–14.

Page 35 of 41

Project Tiger. (2005). The report of the Tiger Task Force: Joining the dots. New Delhi: Union Ministry of Environment and Forests.

Raymakers, C. (2002). Study on the social and economic aspects of illegal fishing in the Caspian Sea. TRAFFIC Europe: WWF.

Reeve, R. (2002). *Policing International Trade in Endangered Species: The CITES Treaty and Compliance*. London: Earthscan Ltd.

Regueira, R. F. S., and Bernard, E. (2012). Wildlife sinks: Quantifying the impact of illegal bird trade in street markets in Brazil. *Biological Conservation*, 149(1), 16-22.

Robbins, C. S. (2003). *Prickly trade: Trade and conservation of Chihuahuan desert cacti.* TRAFFIC North America: WWF.

Roe, D. (2002). Making a killing or making a living: Wildlife trade, trade controls, and rural livelihoods (No. 6). IIED.

Roe, D. (2008). Trading nature: A report, with case studies, on the contribution of wildlife trade management to sustainable livelihoods and the Millennium Development Goals. TRAFFIC international.

Roe, D., Mulliken, T., Milledge, S., Mremi, J., Mosha, S., and Grieg-Gran, M. (2002). *Making a killing or making a living: Wildlife trade, trade controls and rural livelihoods*. London: IIED, and Cambridge: TRAFFIC.

Rosen, G. E., and Smith, K. F. (2010). Summarizing the evidence on the international trade in illegal wildlife. *EcoHealth*, 7(1), 24–32.

Rosser, A. M., and Mainka, S. A. (2002). Overexploitation and species extinctions. *Conservation Biology*, *16*(3), 584–586.

Saffron, I. (2002). Caviar: The strange history and uncertain future of the world's most coveted delicacy. New York: Broadway Books.

Schneider, J. L. (2008). Reducing the illicit trade in endangered wildlife: The market reduction approach. *Journal of Contemporary Criminal Justice*, 24, 274–295.

Schneider, J. L. (2012). Sold into extinction: The global trade in endangered species. ABC-CLIO.

Sellar, J. M. (2007). International illicit trafficking in wildlife. *The Police Chief*, 74(6), 26–32.

Semiadi, G., Darnaedi, D., and Arief, A. J. (2008). Sunda pangolin manis javanica conservation in Indonesia: Status and problem. TRAFFIC Southeast Asia. Proceedings of

the workshop on trade and conservation of pangolins native to South and Southeast Africa, 12–17.

Shelley, T. O., and Crow, M. S. (2009). The nature and extent of conservation policing: Law enforcement generalists or conservation specialists. *American Journal of Criminal Justice*, 34, 9–27.

Shepherd, C. R. (2009). Overview of pangolin trade in Southeast Asia. TRAFFIC Southeast Asia. Proceedings of the workshop on trade and conservation of pangolins native to South and Southeast Africa, 6–9.

Shepherd, C. R., & Magnus, N. (2004). *Nowhere to hide: the trade in Sumatran Tiger.* TRAFFIC.

Shepherd, C. R., & Shepherd, L. A. (2010). The poaching and trade of Malayan un Bears in Peninsular Malaysia. *TRAFFIC Bulletin*, 23(1), 49–52.

Sherman, L. W., Gartin, P. R., and Buerger, M. E. (1989). Hot spots of predatory crime: Routine activities and the criminology of place*. *Criminology*, *27*(1), 27–56.

Shibao, W., Guangzhi, M., and Qianxiang, L. (2005). Research on the conservation biology of Chinese pangolin Manis pentadactyla. Beijing: China Forestry Publishing House.

Sidebottom, A. (2013). On the application of CRAVED to livestock theft in Malawi. *International Journal of Comparative and Applied Criminal Justice*, *37*(3), 195–212.

Smith, R. J., and Walpole, M. J. (2005). Should conservationists pay more attention to corruption? *Oryx*, 39(3), 251–256.

Speart, J. (1993). War Within. Buzzworm: The Environmental Journal, 5(4), 36-45.

Sterling, E. J., Hurley, M. M., and Le, M. D. (2006). *A natural history*. New Haven, CT: Yale University Press.

Sutherland, W. J., Adams, W. M., et al. 2009. One hundred questions of importance to the conservation of global biological diversity. *Conservation Biology*, 23(3), 557–567.

Sutter, J. (2014). The most trafficked mammal you never heard of. Retrieved Sep 1, 2015, http://www.cnn.com/interactive/2014/04/opinion/sutter-change-the-list-pangolintrafficking/.

Stoddard, E. (2012, Apr 26). Legalising rhino horn, ivory trade in focus. Retrieved Jan 10, 2016, http://www.reuters.com/article/ozatp-africa-money-idAFJOE83P03N20120426.

Stretesky, P. B., Long, M. A. and Lynch, M. J. (2013). *The treadmill of crime: Political economy and green criminology*. New York: Routledge.

Page 37 of 41

Tailby, R., and Gant, F. (2002). The illegal market in Australian abalone. *Trends and Issues in Crime and Criminal Justice*. Canberra: Australian Institute of Criminology.

Tella, J. L., and Hiraldo, F. (2014). Illegal and legal parrot trade shows a long-term, cross-cultural preference for the most attractive species increasing their risk of extinction. *PLoS One*, *9*(9), 1–10.

Tilson, R., Nyhus, P. J., Sriyanto, Rubianto, A. (2010). Poaching and poisoning of tigers in Sumatra for the domestic market. In *Tigers of the world: The science, politics and conservation of Panthera tigris*, ed. R. Tilson, P. J. Nyhus, 2nd ed., 101–112. London: Academic Press.

To A. W. L., Hau, B. C. H. and Lee, S. K. H. (2006). A Study on the Trade in Dried Abalone in Hong Kong. *TRAFFIC Bulletin*, 21(1), 25–34.

't Sas-Rolfes, M. (2000). Assessing CITES: Four case studies. In *Endangered species* threatened convention: The past, present and future of CITES, 69–87. London: Earthscan.

't Sas-Rolfes, M. (2010). Tigers, economics, and the regulation of trade. In *Tigers of the world: The science, politics, and conservation of Panthera tigris*, ed. R. Tilson and P. J. Nyhus, 2nd. ed., 477–490. Amsterdam: Academic Press.

't Sas-Rolfes, M. (2012). The rhino poaching crisis: A market analysis. Retrieved Dec 1, 2015, http://www.rhino-economics.com/publications/.

TRAFFIC. (1997). Rhino progress? The response to CITES resolution conference 9.14: A TRAFFIC network report. June: 1–5.

TRAFFIC. (2008). What's driving the wildlife trade? A review of expert opinion on economic and social drivers of the wildlife trade and trade control efforts in Cambodia, Indonesia, Lao PDR and Vietnam. East Asia and Pacific Region Sustainable Development discussion papers. East Asia and Pacific Region Sustainable Development Department. Washington DC: World Bank.

TRAFFIC. (2015, Jan 28). Transportation and logistics sector lends support to global efforts tackling wildlife crime. Retrieved Jan 2, 2016, http://www.traffic.org/home/2015/1/28/transportation-and-logistics-sector-lends-support-to-global.html.

van Lavieren, E. (2008). The Illegal Trade in Barbary Macaques from Morocco and Its Impact on the Wild Population. *TRAFFIC Bulletin*, 21(3), 123–130.

Vasquez, J. C. (2003). Compliance and enforcement mechanisms of CITES. In *The trade in wildlife regulation for conservation*, ed. S. Oldfield, 63–69. London: Earthscan.

Vince, G. (2002). Organized gangs target wildlife trade. *New Scientist*. Retrieved Sep 1, 2015 at: https://www.newscientist.com/article/dn2413-organised-gangs-target-wildlife-trade/.

Vira, V., Ewing, T. and Miller, J. (2014). *Out of Africa: Mapping the global trade in illicit elephant ivory*. Center for Advanced Defense.

Von Essen, E., Hansen, P. H., Kallstrom, H. N., Peterson, M. N. and Peterson, T. R. (2014). Deconstructing the poaching phenomenon: A review of typologies for understanding illegal hunting. *The British Journal of Criminology*, *54*, 632–651.

Wang, G. B. (2008). Pangolin conservation in Taiwan. TRAFFIC Southeast Asia. Proceedings of the workshop on trade and conservation of pangolins native to South and Southeast Africa, 80–83.

Warchol, G. (2004). The transnational illegal wildlife trade. *Criminal Justice Studies*, 17, 57-73.

Warchol, G. and Kapla, D. (2012). Policing the wilderness: A descriptive study of wildlife conservation officers in South Africa. *International Journal of Comparative and Applied Criminal Justice*, *36*, 83–101.

Warchol, G., Zupan, L., and Clack, W. (2003). Transnational Criminality: An Analysis of the Illegal Wildlife Market in Southern Africa. International. *Criminal Justice Review*, 13, 1–27.

Weber, D. S., Mandler, T., Dyck, M., De Groot, P. J. V. C., Lee, D. S., and Clark, D. A. (2015). Unexpected and undesired conservation outcomes of wildlife trade bans—An emerging problem for stakeholders?. *Global Ecology and Conservation*, *3*, 389-400.

Weston, M. K., & Memon, M. A. (2009). The illegal parrot trade in Latin America and its consequences to parrot nutrition, health and conservation. *Bird populations*, *9*, 76–83.

White, R. (2008). *Crimes against nature: Environmental criminology and ecological justice*. Portland, OR: Willan.

White, R. (2011). *Transnational Environmental Crime: Toward an eco-global criminology.* New York: Routledge.

White, R. and Heckenberg, D. (2014). *Green criminology: An introduction to the study of environmental harm*. New York: Routledge.

Wilson-Wilde, L. (2010). Wildlife crime: A global problem. *Forensic Science, Medicine, and Pathology, 6*(3), 221–222.

Wright, T. F., Toft, C. A., Enkerlin-Hoeflich, E., Gonzalez-Elizondo, J., Albornoz, M., Rodríguez-Ferraro, A., and Brice, A. T. (2001). Nest poaching in neotropical parrots. *Conservation Biology*, 15(3), 710–720.

Wu, S., Liu, N., Zhang, Y., and Ma, G. Z. (2004). Assessment of threatened status of Chinese Pangolin (Manis pentadactyla). *Chinese Journal of Applied and Environmental Biology*, 10(4), 456-461.

Page 39 of 41

WWF (2015). Greater one-horned rhino. Retrieved Dec 1 2015 from: http://www.worldwildlife.org/species/greater-one-horned-rhino.

WWF. (2012). Fighting illicit wildlife trafficking: A consultation with governments. Gland, CH: WWF International.

Wyatt, T (2009). Exploring the organization of Russia Far East's illegal wildlife trade: Two case studies of the illegal fur and illegal falcon trades. *Global Crime*, 10(1-2), 144-154.

Wyatt, T. (2013). Wildlife trafficking: A deconstruction of the crime, the victims and the offenders. New York: Palgrave Macmillan.

Yue, Z. (2009). Conservation and trade control of pangolins in China. Workshop on trade and conservation of pangolins native to South and Southeast Asia, 66.

Zimmerman, ME (2003). The black market for wildlife: Combating transnational organized crime in the illegal wildlife trade. *Vanderbilt Journal of Transnational Law*, 36, 1657–1689.

Notes:

- (1) Retrieved on January 17. 2016 https://cites.org/eng/news/pr/2013/20130523_un_lra.php. A recent *National Geographic* also investigated this potential link: http://www.nationalgeographic.com/tracking-ivory/article.html.
- (2) Pseudo-hunting comes in the form of nonhunters, such as Vietnamese citizens, getting hunting permits in order to obtain rhino horn and take it home legally. Many of such hunters had to be trained to shoot a gun and, sometimes, had the guide shoot the rhino on their behalf. Between 2009 and 2012, nearly half of all foreign permit holders in South Africa were from Vietnam (Milliken and Shaw 2012).

Stephen F. Pires

Assistant Professor, Department of Criminal Justice, Florida International University

William D. Moreto

Assistant Professor, Department of Criminal Justice, University of Central Florida

